# DIGITALES ARCHIV

ZBW – Leibniz-Informationszentrum Wirtschaft ZBW – Leibniz Information Centre for Economics

Khokhar, Abdul-Rahman

Article Working capital investment : a comparative study : Canada versus the United States

Multinational finance journal

**Provided in Cooperation with:** Multinational Finance Society

*Reference:* Khokhar, Abdul-Rahman (2019). Working capital investment : a comparative study : Canada versus the United States. In: Multinational finance journal 23 (1/2), S. 65 - 102. http://www.mfsociety.org/modules/modDashboard/uploadFiles/journals/ MJ~0~p1dm87fs3914rs16fi136ink34gn4.pdf.

This Version is available at: http://hdl.handle.net/11159/5509

Kontakt/Contact ZBW – Leibniz-Informationszentrum Wirtschaft/Leibniz Information Centre for Economics Düsternbrooker Weg 120 24105 Kiel (Germany) E-Mail: *rights[at]zbw.eu* https://www.zbw.eu/econis-archiv/

#### Standard-Nutzungsbedingungen:

Dieses Dokument darf zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden. Sie dürfen dieses Dokument nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, aufführen, vertreiben oder anderweitig nutzen. Sofern für das Dokument eine Open-Content-Lizenz verwendet wurde, so gelten abweichend von diesen Nutzungsbedingungen die in der Lizenz gewährten Nutzungsrechte.



BY NC https://zbw.eu/econis-archiv/termsofuse

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

#### Terms of use:

This document may be saved and copied for your personal and scholarly purposes. You are not to copy it for public or commercial purposes, to exhibit the document in public, to perform, distribute or otherwise use the document in public. If the document is made available under a Creative Commons Licence you may exercise further usage rights as specified in the licence.



## Working Capital Investment: A Comparative Study – Canada Versus the United States

Abdul-Rahman Khokhar\* Saint Mary's University, Canada

This study empirically compares the working capital investment of industrial firms and finds that Canadian firms invest less in working capital than their U.S. counterparts. Matched samples of 8,628 firm-year observations each from Canada and the U.S. are utilized covering the period 1988 to 2016. Compared to their U.S. counterparts, Canadian firms have a significantly lower cash conversion cycle, non-cash working capital to asset ratio and non-cash working capital to sales ratio. The difference in working capital investment is robust to variety of firm, industry and country controls as well as to year and industry fixed effects. The study also investigates the determinants of the lower investment in working capital by Canadian firms and finds that working capital investment is negatively moderated by short-term interest rates and positively associated with international operations. (JEL: G30, G31, G32)

- Keywords: working capital management; cash conversion cycle; working capital investment; short-term financial policies
- Article history: Received: 14 May 2018, Received in final revised form: 26 February 2019, Accepted: 17 May 2019, Available online: 3 October 2019

#### I. Introduction

Decision-making in the corporate finance field is broadly divided into long-term (capital budgeting and capital structure) and short-term (working capital management) activities. Traditionally, capital

<sup>\*</sup> Tel.:+1-902-491-6371; Fax:+1-902-496-8101. E-mail address: Rahman.Khokhar@smu.ca. We thank Dr. Trevor Chamberlain. Dr. Mohammad Rahaman, Dr. Sudipto Sarkar and seminar participants at Sobey School of Business for their valuable feedback. We also acknowledge the valuable comments from three anonymous referees and the Editor.

<sup>(</sup>Multinational Finance Journal, 2019, vol. 23, no. 1/2, pp. 65–102) © Multinational Finance Society, a nonprofit corporation. All rights reserved.

#### **Multinational Finance Journal**

budgeting and capital structure decisions have received more attention from researchers than working-capital management (Ramiah et al., 2016) notwithstanding the latter being an essential factor to a firm's smooth operation and survival (Smith, 1979). Despite the importance and implications of working capital policies, the literature provides no comparative evidence on such policies. To the best of the author's knowledge, no published study has compared working capital policies of Canadian and U.S. firms. The objective of the paper is to fill the void in the literature by empirically testing for any cross-country differences in working capital investment, as well as determinants of such differences, if any.

Given the fact that multiple firm-, industry- and country-level factors influence working capital decisions, it is challenging to select countries for the purpose of comparative analysis. Selecting countries with substantially different formal and informal institutions and which are geographically distant (such as Canada versus Australia or the UK) may lead to spurious empirical identification. Instead, a comparison of firms' working capital policies between Canada and the U.S. is more appropriate. For instance, geographic proximity of these two countries mitigates concerns that comparison countries are very different from each other with respect to formal and informal institutions. On the other hand, the differences in cultural, firm and country characteristics between Canada and the U.S. provide enough heterogeneity for a robust empirical comparison (Khoury, Smith and MacKay, 1999). More precisely, the aim is to answer the following two questions: 1) Is working capital investment of Canadian firms different from that of U.S. firms as reported by the survey studies?; 2) What are the determinants of working capital investment differences, if any, between Canada and the U.S.?

Optimal working capital is determined through the trade-off between the cost of financing and benefits of investing in working capital. The study predicts that given the positive effects of working capital on firm value and operating profits, as discussed above, a typical firm in Canada or the U.S. has an incentive to invest more in working capital as long as its marginal benefits outweigh the marginal costs. It is argued that the optimal working capital investment will fluctuate at the firm level, irrespective of whether it differs at the country level. This notion is supported by the survey research, which finds both differences and similarities among Canada, the U.S. and Australia (Belt and Smith, 1991; Smith and Belt, 1989; Smith, 1979). Whether or not cross-country working capital investment is different between Canada and the U.S. is an empirical question explored in this study.

Since working capital accounts such as inventory, receivables and payables are linked to each other (Kieschnick, Laplante and Moussawi, 2013), integrated measures are used, rather than individual proxies, for working capital investment. These include the cash conversion cycle (CCC), the non-cash working capital to asset ratio (WAR). The main independent variable is an indicator variable for Canadian firms (*CANADA*) that equals 1 for a firm-year observation of a Canadian firm and 0 for a U.S. firm.

The analysis relies on matched samples of 8,628 firm-year observations each from Canada and the U.S. for the period 1988 to 2016. First, compared to the U.S. sample, both the mean and median working capital investment of Canadian firms are significantly lower across three alternative proxies. For instance, the mean CCC, WSR and WAR are 58.06 (87.82), 0.17 (0.21) and 0.17 (0.23), respectively, for the Canadian (U.S.) sample. Then conditional tests are conducted while controlling for a number of firm, industry and country factors to confirm whether the lower working capital investment of Canadian firms is robust in a multivariable setting. Ordinary least squares and quintile regressions with year and firm fixed effects and firm-level clustering find significantly lower working capital investment by Canadian firms. The differences in mean working capital investment are economically significant as the WAR and WSR difference represent \$117.94 million of total assets and \$118.12 of total sales, respectively, for an average firm in the sample.

Additionally, since the main analysis concludes that working capital investment is different across Canadian and U.S. firms, firm- and country-level determinants of such differences are explored. Two possible channels are proposed to explain lower working capital investment by Canadian firms relative to U.S. firms. On one hand, the firm-level higher short-term interest cost, on average, would encourage efficient investment in working capital leading to a lower level of working capital investment. Since it is difficult to observe the actual firm-level short-term interest rates for such a large sample due to non-availability of data, an indirect method is applied by using the country-level 3-month T-Bill rate as a proxy for interest rate cost. In other words, the prediction is that higher short-term interest rates moderate the difference in working capital investment between

#### **Multinational Finance Journal**

Canadian and U.S. firms. It is argued that the smaller size of the Canadian economy limits the size of aggregate supply of short-term finance and hence leads to relatively higher short-term interest rates for an average firm in Canada. This argument is further supported by the fact that, compared to the U.S. firms, Canadian firms are smaller and younger, which indirectly suggests that Canadian firms, on average, may pay a higher premium over a country-level policy interest rate such as T-bill rate or bank prime rate.<sup>1</sup> The highly integrated nature of some industry sectors in Canada and the U.S. raises a concern that some Canadian firms may borrow short-term funds in U.S. dollars using a U.S. benchmark policy rate. While this limitation is acknowledged, it is assumed that a much larger proportion of Canadian firms borrow mostly in Canadian dollars and, hence, it is expected that this concern will not qualitatively change the findings.

On the other hand, at the firm level, it is suggested that relatively lower dependence of Canadian firms vis-à-vis U.S. firms on foreign sales is a second channel for lower investment in working capital. It is difficult to predict the impact of international operations on working capital differences between Canada and the U.S. Depending on the specific nature of the firm's operations, it could either increase or decrease optimal working capital investment. For instance, firms with high demand may decide to produce in foreign countries to reduce working capital investment. Alternatively, firms with limited foreign demand may find it optimal to export their products from the home country, which would increase investment in working capital. There is anecdotal evidence reported by CFO magazine that multinational firms invest more in inventories because of international sourcing (Over a Barrel, 2016) or take longer to collect cash because of greater shipping time for intermediate goods between countries.<sup>2</sup> Specifically, the effect of international operations on cross-country differences in working capital investment is investigated as an empirical question.

2. An anonymous referee provided this anecdotal evidence.

68

<sup>1.</sup> It is argued that bank prime rate may be a better proxy for short-term interest rates, especially for smaller firms that cannot access the money market. Therefore, the 3-months' T-bill rate is replaced with the bank prime rate to measure country-level short-term interest rates and the results, available upon request, are unchanged. In addition, as pointed out by an anonymous referee, a shift is observed in relative prime rates between Canada and the U.S. during the mid-1990s. To mitigate this concern, the main result in table 4 is re-estimated using two time-series sub-samples (1988-1997 and 1998-2016) and shows qualitatively and quantitatively similar results (not reported for brevity).

Mean and median regression analyses are used and introduce interaction terms between CANADA and the two proposed channels (short-term interest rates and international operations) in addition to control variables, year and industry fixed effects and firm-level clustering. Consistent with expectations, in the mean regressions, negative coefficients are found for the interaction terms between CANADA and short-term interest rates for the CCC measure (-1.1390 though insignificant), WAR (-0.0046 significant at the 1% level) and WSR (-0.0034 insignificant). In the median regressions, these interaction coefficients are -0.7273 (insignificant), -0.0064 (significant at the 1% level) and -0.0056 (significant at the 1% level). With respect to the interaction term between CANADA and the international dummy, all coefficients are positive and significant at the 1% level in both mean and median regressions. The coefficients for mean (median) regression are 26.6658 (22.0091) for CCC, 0.0466 (0.0483) for WSR and 0.0544 (0.0496) for WAR. These results confirm that lower working capital investment by Canadian firms is moderated by higher short-term interest rates and a lower percentage of international firms operating in Canada.

The robustness of the findings is verified by conducting several complementary tests. Specifically, industry-adjusted values are used for three dependent variables and similar results are found in both mean and median regressions. Next, given a large number of independent variables, Leamer's (1985) extreme bounds analysis (EBA) is used to determine which independent variables are robust and finds that the results are unchanged after excluding the fragile explanatory variables in the multivariable analysis. Then alternative methods are used to calculate the length of the cash conversion cycle (CCC) by replacing average working capital amounts with year-end to compute and test the multivariable specification with this alternative CCC measure. In another robustness check, the difference between the working capital measures in the two countries is used as dependent variables and the results are generally consistent with the findings with respect to the determinants of working capital investment across Canada and the U.S. Moreover, multivariable models are estimated using the raw unmatched sample instead of the matched sample, with 9,602 and 67,820 firm-year observations for Canada and the U.S., respectively. In addition, propensity score matching technique (PSM) based on firm size, leverage and market-to-book is used as an alternative criterion to match the samples in order to mitigate any variation in working capital investment across industries. Finally, the control variables set is supplemented by

including a profitability measure (return on assets) in the conditional analysis. The results for all the robustness tests, though not reported for brevity, are similar to the main findings.

This study contributes to the extant corporate finance literature in three distinct ways. First, it augments the prior survey research on cross-country comparisons of working capital policies (Khoury, Smith and MacKay, 1999; Belt and Smith, 1991; Smith, 1979) across Australia, Canada and the U.S. These studies document similarities and differences in working-capital policies across countries and across time. Whereas, this paper provides a first statistical comparison of working capital investment and complements the survey evidence with respect to the difference in working capital policies. Firm working capital investment is found to be associated with the country-level factors such as short-term interest rates and whether or not the firm operates abroad. Second, this paper is related to the research stream that explores determinants of working capital policies such as governance, cash flows, public versus private ownership and national culture (Zeidan and Shapir, 2017; Ben-Nasr, 2016; Elo and Tanska, 2016; Gill and Biger, 2013; Anagnostopoulou, 2013; Chiou, Cheng and Wu, 2006). This line of research is extended by identifying two additional determinants, namely short-term interest rates and international operations, which influence the level of working capital investment. Finally, the study provides indirect support to the research exploring the relationship between working capital management and firm value. For instance, Aktas, Croci and Petmezas (2015) argue that working capital investment can have opposing effects on firm value conditional on the net cost-benefit trade-off of investing in individual working capital accounts such as receivables, inventories and payables and report a non-linear relationship between working capital investment and firm value. This paper supplements their results by showing that country-level characteristics (short-term interest rates) and operational structure (international versus domestic) affect the level of working capital investment that maximizes firm value.

The rest of the paper is organized as follows. Section II provides a literature review and Section III discusses sample construction, variables and empirical prediction. Section IV presents the results of comparative analysis and Section V explores the determinants of the difference in working capital investment between Canada and the U.S. Section VI describes the results of the additional tests to confirm the robustness of the analysis. Section VII concludes the study.

#### **II.** Literature review

Working capital management involves "day-to-day activity that ensures the firm has sufficient resources to continue the operations and avoid costly interruptions" (Ross et al., 2016: 5) and is important for several other reasons. First and foremost is the fact that corporate firms invest a significant amount in working capital. For example, U.S. firms' had \$4.2 trillion invested in working capital at the end of 2011 (Aktas, Croci and Petmezas, 2015). Similarly, during the period 1988 to 2016, an average Canadian (U.S.) firm held 17% (21%) of total sales or 17% (23%) of total assets in non-cash working capital. Working-capital management is also believed to have an impact on a firm's capital structure (Chiou, Cheng and Wu, 2006), suggesting that long-term financial decisions are related to short-term financial decisions. It is further argued that working capital can be used as a backup resource to avoid default (Bierman, Chopra and Thomas, 1975), cash substitute (Bates, Kahle and Stulz, 2009), and alternative source of financing (Fazzari and Petersen, 1993; Eckbo and Kisser, 2013). The extant literature also provides evidence that efficient working capital management has implications for firm value (Albuquerque, Ramadorai and Watugala, 2015; Kieschnick, Laplante and Moussawi, 2013; Almeida and Eid, 2014; Zeidan and Shapir, 2017), profitability (Shin and Soenen, 1998; Deloof, 2003; Lazaridis and Tryfonidis, 2006; García-Teruel and Martinez-Solano, 2007), operating performance (Zeidan and Shapir, 2017; Box et al., 2018), cash flows (Zeidan and Shapir, 2017), governance (Ben-Nasr, 2016; Gill and Biger, 2013), private versus listed firms (Anagnostopoulou, 2013) and national culture (Elo and Tanska, 2016).

Notwithstanding the aforementioned importance, there has been very little research on the comparison, especially statistical, of working capital investment across countries. The only exceptions are surveys comparing working capital policies of Canadian and U.S. firms. For instance, Khoury, Smith and MacKay (1999) surveyed financial executives to compare working-capital practices in Canada, the U.S. and Australia and found similarities and differences across countries as well as across time. In another survey, Belt and Smith (1991) attempted to understand the working capital management for a sample of Australian firms by comparing the results with two previous studies in the U.S. (Smith and Belt, 1989; Smith, 1979), and reported both similarities and differences. This paper argues that survey instruments fail to

#### **Multinational Finance Journal**

appropriately identify cross-country differences in working capital policies for a number of reasons. First, survey results might be biased because there may be differences between what managers say in survey responses and what they actually do (Graham, 2004). Second, the survey methodology is most appropriate for a setting where information or data are unavailable for the population and one needs to generalize the conclusion by exploring a subset of the population (Baker and Mukherjee, 2007; Baker, Dutta and Saadi, 2011). With regard to working capital investment, the data are readily available for publicly listed firms, especially in advanced economies. Moreover, the survey evidence to date has been mixed, suggesting both similarities and differences between Canada and the U.S. Finally, compared to the U.S., the robustness of the Canadian economy during and after the 2008 financial crisis suggests that Canadian firms may behave differently when it comes to managing working capital investment. Taken together, it is important to investigate the cross-country statistical comparison of working capital investment policies.

Finally, the study is relevant to the research stream that explores determinants of investment in working capital. Thus far, there is evidence for a relationship between working capital investment and several factors including firms size, leverage, firm age, fixed asset growth, cash flows, book-to-market, profitability, interest rates, intangible assets, research and development expenses, firm-specific risk, cash reserves, variation in sales, sales growth, ownership structure and industry (Anagnostopoulou, 2013; Aktas, Croci and Petmezas, 2015; Baños-Caballero, García-Teruel and Martínez-Solano, 2010; Ben-Nasr, 2016; Chiou, Cheng and Wu, 2006; Ding, Guariglia and Knight, 2013; Hawawini, Viallet and Vora, 1986; Kieschnich, Laplante and Moussawi, 2006; Shin and Soenen, 1998; Weinraub and Visscher, 1998).<sup>3</sup>

#### III. Sample, variables and empirical predictions

#### A. Sample construction

The financial statement data for all listed Canadian and U.S. firms are sourced from Standard & Poor's Compustat fundamental annual files

<sup>3.</sup> Aktas, Croci and Petmezas (2015) provide a list of factors associated with working capital investment decisions.

for the fiscal years 1988 to 2016.<sup>4,5</sup> Consistent with the extant literature, the focus is on the working capital investments of industrial firms, with financial firms and utilities (Standardized Industry Classification (SIC) Codes 6000 to 6999 and 4900 to 4999) excluded. Firm observations with missing data for fiscal year and stock price are deleted, as are the observations with zero or negative total assets, current assets, current liabilities, receivables and sales. All variables are annual and winsorized at the 1% tails to suppress outliers. This strategy yields a sample of 9,602 and 67,820 firm-year observations for Canada and the U.S., respectively. Given a significant difference in number of firm-year observations between the two countries, the comparison of working capital investment may be biased. It is important to create a matching sample to ensure an unbiased comparison of working capital investment between the two countries. In order to create a matching sample, the PSM technique is used to create a matching nearest neighbor U.S. firm-year observation for each Canadian firm-year observation based on firm size and 2-digit SIC code. This methodology finds a matched pair of cross-country firms having similar firm size and industry segment for a robust comparison. The final sample consists of 8,628 firm-year observations each from Canada and the U.S. over the period 1988 to  $2016.^{6}$ 

#### B. Variables

#### Main variables

The literature has used both individual and integrated measures as

<sup>4.</sup> This study uses a common data source for Canadian and U.S. firms to maximize number of observations. Another source for Canadian firms' data is the Financial Post, which annually publishes a list of the Top 500 Canadian firms. The Compustat sample is manually compared with the 2017 FP 500 list to ensure completeness of the Canadian sample; over half of the FP500 Canadian firms were present in the Compustat sample. Several FP500 Canadian firms not present in the sample either belong to the financial/utilities sector (excluded in this study) or are too small to make the FP500 list.

<sup>5.</sup> The sample period is constrained due to cash flow statement data starting in 1986 in Compustat.

<sup>6.</sup> In total, there are 564 and 3,246 unique firms from Canada and the U.S., respectively. The lower number of unique Canadian firms in the matched sample is explained by the fact that various yearly observations of one Canadian firm are matched with different U.S. firms across the time period 1988 to 2016. For example, Canadian firm X could match with U.S. firm Y for fiscal year 2000 and U.S. firm Z for fiscal year 2001.

#### **Multinational Finance Journal**

proxies for working capital investment. Petersen and Rajan (1997), Box et al. (2018), and Albuquerque, Croci and Petmezas (2015) explore working capital management by studying individual accounts such as receivables and trade credit. Other studies (Kieschnick, Laplante and Moussawi, 2013; Aktas, Croci and Petmezas, 2015; Ben-Nasr, 2016; Ding, Guariglia and Knight, 2013) have used an integrated measure of non-cash working capital, that is, inventories plus receivables minus accounts payables. Kieschnick, Laplante and Moussawi (2013) argue that the management of working capital accounts cannot be studied in isolation since these accounts are fundamentally linked to each other. For example, a firm's generous trade credit policy would not only result in higher sales and receivables but also necessitate a higher investment in inventory. They suggest using the cash conversion cycle as a measure of net working capital investment because it jointly captures the management of receivables, inventory and payables. Similarly, Aktas, Croci and Petmezas (2015) and Ben-Nasr (2016) use the net working capital to sales ratio and Ding et al. (2013) use the cash conversion cycle as measures for working capital investment. Consequently, for the dependent variable, three integrated proxies are used for working capital investment: the cash conversion cycle (CCC), the net working capital to asset ratio (WAR) and the net working capital to sales ratio (WSR).<sup>7</sup> Since the aim is to discover whether Canadian firms invest differently in working capital than U.S. firms, a Canadian country dummy (CANADA) is used as the main independent variable, equal to 1 if the firm-year observation belongs to Canada and 0 otherwise. The variable definitions are provided in the appendix.

#### Control variables

Working capital research has shown that several firm-level factors influence working capital policies. It is, therefore, essential to control for such factors to ensure unbiased estimations of country effects on such policies. A recent study by Aktas, Croci and Petmezas (2015), which explores whether working capital management is value-enhancing, provides a set of controls. This study follows theirs and controls for firm size (SIZE), intangible assets (INTANGIBLES), leverage (LEVERAGE), firm age (AGE), research and development

<sup>7.</sup> In addition, the study compares the scaled proxies for individual working capital investment measures including inventories (INV), payables (A/P) and receivables (A/R) as robustness checks.

expenses (R&D), firm-specific risk (RISK), growth in fixed asset (GFA), cash reserves (CASH RESERVE),<sup>8</sup> variation in sales (SALES VOLATILITY), cash flows (CF), and sales growth (SALES GROWTH). Given the cross-country comparison in this study, it is important to control for country-level heterogeneity for unbiased identification. First, since firms predominantly finance working capital through short-term borrowings or revolving lines of credit, a country-level difference in short-term interest rates may affect working capital policies. In addition, working capital investment can be affected if a firm operates in multiple countries as opposed to only in the domestic market. Therefore, a control for short-term interest rates (STIR) and an international dummy (INTERNATIONAL) are implemented. Working capital requirements for a typical firm may vary over time as well as across industries (Hawawini, Viallet and Vora, 1986; Weinraub and Visscher, 1998). To ensure robust identification, industry and year fixed effects are included in the multivariate specifications.9 Finally, since few country-level variables are included that do not vary across firms, it is not possible to include firm fixed effects in this study. However, firm-level clustering is used to mitigate the possibility of unobserved firm fixed effects.

#### C. Empirical predictions

The literature about cross-country differences in working capital investment is limited to survey studies, the evidence from which is mixed. However, the within-country research, discussed in Section I, identifies several factors including capital structure, cash flows, governance structure and culture as important. Working capital management is a trade-off between the benefits associated with working capital investment and the corresponding costs to finance working

<sup>8.</sup> Since cash reserves may be correlated to working capital measures, especially cash conversion cycle, cash reserves are excluded in a robustness check and similar results are found.

<sup>9.</sup> Accounting literature (Weiner, 2005; Bhojraj, Lee and Oler, 2003; Krishnan and Press, 2003) debates the weakness of Standard Industry Classification (SIC) codes to classify industries. Consequently, this study uses three alternative measures, namely, North American Industry Classification System (NAICS), Global Industry Classifications Standard (GICS) and Fama–French 48 Industry Segments, to check the robustness of findings. Specifically, table 4 is estimated using each of these industry measures and results are quantitatively and qualitatively similar (not reported for brevity).

capital. With no financing constraints, a firm will continue to increase investment in working capital if the marginal benefit of additional investment exceeds the marginal cost of additional financing. In theory, all firms should achieve an optimal level of working capital investment. The firm-level or, aggregately, the country-level optimal working capital investment will depend on the firm's cost of financing or the expected benefits from the working capital investment. However, for two reasons it is difficult to predict a priori whether working capital investment of Canadian and U.S. firms is different or similar. One, prior research lacks an evidence-based statistical comparison of cross-country working capital investment. Second, survey studies (Khoury, Smith and MacKay, 1999; Belt and Smith, 1991; Smith and Belt, 1989; Smith, 1979) provide mixed evidence with respect to cross-country differences and/or similarities. Consequently, whether working capital investment is similar or different across Canadian and U.S. firms is an open question best resolved through empirical analysis. Additionally, if the main analysis concludes that working capital investment is different across Canadian and U.S. firms, the firm- and country-level factors explaining such differences will be explored.

### IV. Comparison of working capital investment: Canada versus U.S.

The study carries out a broad analysis to investigate similarities or differences in working capital investment between Canadian and U.S. firms. First, the study checks whether the CCC, WSR, WAR, INV, A/P and A/R of Canadian firms are statistically different from those of U.S. firms. The results of univariate differences between the working capital investments of matched Canadian and U.S. firms are reported in table 1, which provides summary statistics including mean, median, minimum and maximum values and number of observations for working capital proxies as well as the control variables.

Lower levels of working capital held by the Canadian firms are observed regardless of the choice of working capital measure. For instance, the mean (median) cash conversion cycle of Canadian firms is 58.05 (54.82) days as opposed to 87.86 (76.74) days for U.S. firms. Similarly, the mean working capital to sales (assets) ratio for the Canadian firms is 17% (17%) compared to 21% (23%) for U.S. firms. The results are similar when individual components of working capital are compared because Canadian firms have, on average, 3% lower cash and 5% higher payables except for inventories and receivables, which are similar across the two countries. Two possible explanations are proposed for the higher payable ratio for Canadian firms. One, given higher short-term interest rates, Canadian firms use trade credit as an alternative source of short-term financing (Ge and Qiu, 2007). Second, a higher payables ratio among Canadian firms is perhaps a result of relatively more use of short-term paper, receivables factoring and/or asset-based lending in the U.S.<sup>10</sup> However, median inventories are 1% lower for Canadian firms. The other interesting differences are in short-term interest rates, where mean is 17 basis points higher and median is 62 basis points lower for Canada, and the international dummy, which shows that only 7% of Canadian firms report foreign income as opposed to 34% of U.S. firms. Since this study uses a matched sample of Canadian and U.S. firms based on firm size, this measure is approximately similar across the countries. Time-series plots are created of the integrated and individual measures of working capital investment as shown in figures 1 and 2, respectively.

The graphs of the integrated working capital measures generally confirm the differences between Canada and the U.S. and show one interesting trend: the gap between Canadian and U.S. working capital investment has narrowed since the 2008 financial crisis. This could be explained, at least in part, by post-crisis changes in short-term interest rates, which reached record lows in Canada, resulting in a smaller gap between Canada and the U.S. The mean plot of individual measures shows trends similar to the integrated measures with a notable difference in payables. The time-series plots provide initial evidence of working capital investment difference and that Canadian firms seem to invest less in working capital than U.S. firms.

Next, to test the statistical significance of the differences in working capital measures in Canada and the U.S., mean and median difference tests are conducted. Table 2 reports these results.

The results in this table confirm the mean (median) differences in working capital investments because, relative to U.S. firms, Canadian firms have a 29.804- (21.916-) day shorter cash conversion cycle, a 3.5% (3.2%) lower working capital to sales ratio and a 5.4% (6.0%)

<sup>10.</sup> Details about factoring or asset-based financing are not observed in these data. However, the main results are robust when the working capital measures are adjusted using the accounts payables net of notes payables-short-term borrowings, which reports use of commercial paper.

TABLE 1. Comparative Summan	ry Statistics						
Variable		Obs	Mean	Median	Std. Dev.	Min	Max
CCC	Canada	8,628	58.05	54.82	111.79	(370.18)	533.28
d S/M	US	8,628 0 670	87.86	76.74	100.34	(370.18)	533.28 1.02
NGW	US	0,020 8.628	0.21	0.10	0.20	(0.54) (0.54)	1.03
WAR	Canada	8,628	0.17	0.15	0.17	(0.20)	0.71
	NS	8,628	0.23	0.21	0.17	(0.20)	0.71
CASH	Canada	8,628	0.11	0.05	0.15	0.00	0.79
	NS	8,628	0.14	0.07	0.17	0.00	0.79
INV	Canada	8,628	0.16	0.12	0.13	0.00	0.64
	NS	8,628	0.16	0.13	0.13	0.00	0.64
A/P	Canada	8,628	0.14	0.12	0.11	0.01	0.56
	NS	8,628	0.09	0.07	0.08	0.01	0.56
A/R	Canada	8,628	0.16	0.14	0.12	0.01	0.55
	NS	8,628	0.16	0.14	0.11	0.01	0.55
SIZE	Canada	8,628	5.63	5.63	2.05	(0.14)	10.15
	NS	8,628	5.75	5.84	2.28	(0.14)	10.15
INTANGIBLES	Canada	8,417	0.10	0.03	0.15	0.00	0.91
	NS	7,851	0.12	0.06	0.16	0.00	0.90
LEVERAGE	Canada	8,628	0.24	0.22	0.22	0.00	1.36
	SU	8,628	0.24	0.22	0.22	0.00	1.36
AGE	Canada	1,984	20.43	19.78	4.37	9.47	33.41
	NS	3,393	23.81	23.61	5.84	3.16	49.41
R&D	Canada	8,628	0.07		0.29	0.00	2.29
	NS	8,628	0.05	0.00	0.18	0.00	2.29
Risk - CFV	Canada	7,782	0.98	0.17	2.88	0.01	21.62
	SU	8,109	0.98	0.20	2.59	0.01	21.62
		( Cont	inued )				

S Su .<u>-</u> చి

78

**Multinational Finance Journal** 

TABLE 1. (Continued)							
Variable		Obs	Mean	Median	Std. Dev.	Min	Max
FIXED ASSET GROWTH	Canada	8,148	0.14	0.04	0.47	(0.64)	3.37
	NS	8,318	0.14	0.04	0.47	(0.64)	3.37
SALES VOLATILITY	Canada	7,782	0.23	0.17	0.19	0.02	1.21
	NS	8,109	0.24	0.18	0.20	0.02	1.21
CF	Canada	8,628	(0.35)	0.20	3.85	(33.74)	6.05
	NS	8,628	(0.16)	0.26	3.54	(33.74)	6.05
BTM	Canada	8,623	0.76	0.62	0.96	(4.46)	4.53
	NS	8,627	0.57	0.47	0.82	(4.46)	4.53
SALES GROWTH	Canada	8,162	0.16	0.07	0.45	(0.59)	2.92
	NS	8,320	0.13	0.07	0.38	(0.59)	2.92
STIR	Canada	8,628	3.43	2.67	2.85	0.20	12.22
	NS	8,628	3.26	3.29	2.50	0.01	8.35
INTERNATIONAL	Canada	8,628	0.07		0.26	0.00	1.00
	NS	8,628	0.34	ı	0.48	0.00	1.00
<b>Note:</b> Comparative summary s are as follows: CCC: cash conversion (DIO) minus days payable outstandii non-cash working capital scaled by asset ratio; A/P: accounts payable tu INTANGIBLES: intangible assets di AGE: number of year since the initi, variation in annual cash flow over th fiscal year; SALES VOLATILITY: extraordinary items and depreciation: SALES GROWTH: change in sales <i>INTERNATIONAL</i> , are winsoriz and maximum values of the variabita and maximum values and the variabita and maximum values of the variabita and maximum values and the variabita and maximum values and the variabita and maximum values and the value and the valu	statistics for Canadi n cycle based on ave mg (DPO) as further the total sales; CAS o the total assets rat ivided by the total as ial public offering; I he past 10 years; FI) coefficient of variat n scaled by the net p over the previous fi le that equals 1 if fir- le that equals 1 if fir- ses.	an and U.S. mat erage values is the explained in the <i>z</i> Hi: cash and mar tio; A/R: account seets; LEVERAG R&D ASSET GR R&D: research a KED ASSET GR tion in annual sal roperty, plant, ar riscal year; <i>STIR</i> : m reports a positi table reports the	ched samples f e sum of the da appendix; WAI ketable securit is receivable to iE: long-term d nd developmer OWTH: change ces to total asser d equipment; J annual percen number of obs	rom 1988 to 20 ys sales outstann ys sales outstann is: non-cash wor tes divided by to the total asset ebt plus debt in it expenses scal in the net prop 3TM: total com age interest rat age interest rat gr income and	16 are presented ding (DSO) and of king capital scale king capital scale is to a sasets; Il ratio; SIZE: natu current liabilities led by the sales; orty, plant and ec ast 10 years; CF mon equity scale es based on 3-mc o otherwise. All v 0 otherwise. All v ns, medians, stan	<ol> <li>The variables days inventory of days inventory of days inventorias NV: inventorias NV: inventorias rral log of the t s divided by the Risk – CFV: co quipment over th resum of inco d by market cap onth Treasury bi variables, except dard deviations</li> </ol>	definitions utstanding sets; WSR: to the total otal assets; otal assets; efficient of he previous inne before italization; Il rate; and age, <i>STIR</i> ,

Working Capital Investment: A Comparative Study – Canada vs US 79



FIGURE 1.— Time Series Comparison of Working Capital Measures – Canada versus U.S.

lower working capital to assets ratio, with all differences significant at the 1% level. The results for the individual working capital components show that, compared to U.S. firms, Canadian firms have 0.4% (0.6%) lower inventories, 5.2% (5.1%) higher payables and 0.1% (0.1%) lower receivables and the differences are again significant at the 1% level, except for receivables where both mean and median differences are insignificant. The comparison of control variables for Canadian and U.S. firms (not reported) indicates significant differences for most variables. In particular, compared to U.S. firms, Canadian firms are younger, have lower cash holdings, intangible assets, leverage, sales volatility and cash flows, and higher research and development expenditures, fixed asset growth, book-to-market ratio, sales growth and interest costs. The results show that, compared to U.S. firms, very few Canadian firms report foreign income. Taken as a whole, this analysis confirms heterogeneity of firm- and country-level characteristics between Canada and the U.S., and suggests the need to control for these factors to ensure robust support for the initial findings.

Spearman rank correlations are computed between the working capital investment proxies and control variables in order to see the unconditional effects of various control variables on working capital investment in each country. The rank correlations for the selected variables are reported in table 3.

First, the study finds that three aggregate measures of working capital investment are highly and significantly correlated; the correlation between working capital measures ranges from 0.65 to 0.91 and all coefficients are significant at the 5% level. Second, with the exception of two variables (short-term interest rates and intangibles), the study finds that the direction of correlation is mostly the same for Canadian and U.S. firms; however, the magnitudes of the correlations vary between the Canadian and U.S. data. The correlation between the cash conversion cycle and short-term interest rate is negative and significant for Canada (-0.05) and positive and significant for the U.S. (0.04). The correlation between the working capital to sales ratio and intangibles is positive and significant for the U.S. (0.433) and negative but insignificant for Canada (-0.0038). Overall, the correlation analysis confirms firm- and country-level heterogeneity among the interactions of control variables and necessitates a conditional analysis for robust identification.

Finally, a conditional analysis is conducted to investigate differences or similarities between the working capital policies of Canadian and





FIGURE 2.— Time Series Comparison of Working Capital Measures – Canada versus U.S.

U.S. firms using the following multivariable specification.

$$WCM_{it} = \beta_0 + \beta_1 CANADA + \beta_2 CONTROLS_{it} + \varepsilon_{it}$$
(1)

where  $WCM_{ii}$  is the year *t* working capital measure for firm *i*, *CANADA* is a dummy for Canadian firms, which equals 1 if firm-year observation belongs to a Canadian firm and 0 otherwise,  $CONTROLS_{ii}$  is the set of control variables for firm *i* in year *t*, and  $\beta_i$  and  $\varepsilon_{ii}$  are model parameters. In this specification, the main parameter of interest is  $\beta_1$ . A significant  $\beta_1$  will suggest that working capital investment of Canadian firms differs from that of U.S. firms, whereas an insignificant  $\beta_1$  will confirm the similarities of working capital investment between these two countries. In addition, the sign of  $\beta_1$ , if significant, will indicate a lower or higher working capital investment across the two countries. For instance, a negative (positive) and significant value for this parameter will indicate that, compared to U.S. firms, Canadian companies invest less (more) in working capital. The study estimates model (1) independently with mean and median values. The results are reported in table 4.

The results confirm significant differences in working capital

#### **Multinational Finance Journal**

investment between Canadian and U.S. firms; the coefficient for the *CANADA* dummy is negative and significant regardless of the choice of working capital investment proxy and the type of regression, suggesting that Canadian firms invest less in working capital. For example, the *CANADA* dummy coefficients in mean and median regressions for CCC are -49.6765 and -28.4566 in Panels I and II, respectively, and both are significant at the 1% level. Similarly, the mean regression coefficient for WSR (WAR) is -0.0571 (-0.0622) and the median regression coefficient is -0.0477 (-0.0425), all significant at the 1% level. The working capital investment differences between Canada and the U.S. are economically significant as well. For example, in the mean regressions, compared to the U.S. sample, the Canadian CCC is 49 days shorter and the WAR and WSR are 5.7% and 6.2% lower, respectively. This translates into values of \$117.94 million in total assets and \$118.12 million in total sales for an average firm in the sample.

The results for the control variables are generally consistent with prior studies (Baños-Caballero, García-Teruel, and Martínez-Solano, 2010; Kieschnich, Laplante and Moussawi, 2013; Chiou, Cheng and Wu, 2006) that have used working capital as the dependent variable to explore the determinants of working capital investment. The study finds that size and leverage are negatively related to working capital investment, while firm age and book-to-market are positively associated. The coefficients for fixed asset growth and short-term interest rate are mixed and mostly insignificant.

Previous empirical studies do not provide the direction of the correlation between working capital investment and cash reserves, intangible assets, research and development, firm-specific risk, sales volatility, sales growth and the international dummy. The results are, however, consistent with theory and intuition as the study finds a negative correlation between working capital investment and cash reserves, intangible assets, research and development expenses, firm-specific risk and sales volatility measures. The correlation between working capital and international dummy is mixed because it is positive for working capital to asset ratio but negative for cash conversion cycle and working capital to sales ratio, whereas it is insignificant between working capital and sales growth.

Taken as a whole, this section strongly confirms differences in working investment and that compared to the U.S., an average firm in Canada invests much less in working capital.

A.					
Variable		Obs	Mean	Difference	P-Value
CCC	Canada	8,628	58.06	29.804***	0.0000
	US	8,628	87.82		
WSR	Canada	8,628	0.17	0.035***	0.0000
	US	8,628	0.21		
WAR	Canada	8,628	0.17	0.054***	0.0000
	US	8,628	0.23		
CASH	Canada	8,628	0.11	0.024***	0.0000
	US	8,628	0.14		
INV	Canada	8,628	0.16	0.004*	-0.0540
	US	8,628	0.16		
A/P	Canada	8,628	0.14	-0.052***	0.0000
	US	8,628	0.09		
A/R	Canada	8,628	0.16	-0.001	-0.4600
	US	8,628	0.16		
A.					
Variable		Obs	Median	Difference	P-Value
CCC	Canada	8,627	54.83	21.916***	0.0000
	US	8,627	76.74		
WSR	Canada	8,627	0.16	0.032***	0.0000
	US	8,627	0.19		
WAR	Canada	8,627	0.15	0.06***	0.0000
	US	8,627	0.21		
CASH	Canada	8,627	0.05	0.022***	0.0000
	US	8,627	0.07		
INV	Canada	8,627	0.12	0.006***	0.0137
	US	8,627	0.13		
A/P	Canada	8,627	0.12	-0.051***	0.0000
	US	8,627	0.07		
A/R	Canada	8,627	0.14	0.001	0.3137
	US	8,627	0.14		

TABLE 2. Mean and Median Difference Tests

**Note:** The table reports the results of mean and median difference tests for working capital measures for Canadian and U.S. matched samples from 1988 to 2016. The variable definitions are provided in table 1. The difference column represents the mean or median value of the U.S. sample minus the mean or median value of the Canadian sample. The significance of the difference in mean values uses the T-test with unequal variance and the significance of difference in median values uses the Mann–Whitney test. The final column reports the p-values. Note that \*\*\*, \*\*, \* show the difference in means or medians is significant at 1%, 5% and 10%, respectively.

TABLE 3. Correlations								
Variables		CCC	WSR	WAR	CASH	INV	A/P	A/R
ccc	Canada	<del></del>						
WSR	US Canada	$0.8981^{*}$	1					
	SU	0.9103*	1					
WAR	Canada	$0.7142^{*}$	0.7478*	1				
	NS	$0.6530^{*}$	$0.6995^{*}$	1				
CASH	Canada	0.0018	0.0039	-0.1697*	1			
	NS	$0.0928^{*}$	0.0335	-0.1332*	1			
INV	Canada	$0.5236^{*}$	$0.4462^{*}$	0.7389*	-0.1853*	1		
	NS	$0.5241^{*}$	0.4795*	$0.7911^{*}$	-0.1349*	1		
A/P	Canada	-0.2324*	-0.2481*	$0.1554^{*}$	-0.1417*	$0.4491^{*}$	1	
	NS	-0.1683*	-0.1298*	$0.3192^{*}$	-0.1906*	$0.5101^{*}$	1	
A/R	Canada	0.1883*	$0.3116^{*}$	$0.5802^{*}$	-0.1015*	$0.3254^{*}$	0.5017*	-
	NS	0.2727*	0.3977*	$0.6826^{*}$	$-0.1091^{*}$	$0.3682^{*}$	0.4745*	1
SIZE	Canada	-0.0584*	-0.0672*	-0.1767*	-0.1706*	-0.2563*	-0.3080*	-0.2643*
	NS	$-0.1456^{*}$	-0.1771*	-0.2746*	-0.2105*	$-0.2516^{*}$	-0.2579*	-0.2770*
INTANGIBLES	Canada	-0.0196	-0.0038	0.0002	-0.0501*	-0.0457	$0.0686^{*}$	$0.1052^{*}$
	NS	0.0254	0.0433*	-0.0837*	$-0.2026^{*}$	-0.1415*	-0.1290*	0.0116
LEVERAGE	Canada	$-0.1586^{*}$	-0.1350*	-0.1091*	-0.4971*	$-0.0491^{*}$	$0.1216^{*}$	$0.0484^{*}$
	NS	-0.1603*	-0.1210*	-0.1013*	-0.5345*	-0.0740*	0.0281	-0.0675*
AGE	Canada	0.0608*	0.0551*	0.0683*	-0.0441	0.0717*	-0.0086	0.0092
	NS	0.0695*	0.0757*	$0.1266^{*}$	$-0.1481^{*}$	0.1115*	0.0801*	$0.0831^{*}$
			( Continu	<i>ed</i> )				

**Multinational Finance Journal** 

86

×								
Variables		CCC	WSR	WAR	CASH	INV	A/P	A/R
R&D	Canada	0.1931*	$0.2311^{*}$	*6960.0	0.4057*	$0.0784^{*}$	0.0878*	0.2246*
	SU	$0.3702^{*}$	0.3555*	$0.0410^{*}$	$0.5130^{*}$	0.0092	-0.1276*	0.0719*
Risk - CFV	Canada	$0.0741^{*}$	$0.1046^{*}$	$0.0764^{*}$	$0.4751^{*}$	$0.1063^{*}$	0.1873*	0.1797*
	NS	$0.2241^{*}$	$0.2262^{*}$	$0.1904^{*}$	0.4572*	$0.1548^{*}$	$0.1402^{*}$	$0.2571^{*}$
FIXED ASSET GROWTH	Canada	0.0271	0.0227	-0.0339	-0.0928*	$-0.1036^{*}$	$-0.1216^{*}$	-0.0775*
	NS	-0.0269	-0.0217	-0.0505*	-0.0948*	-0.0553*	-0.0911*	-0.0729*
SALES VOLATILITY	Canada	-0.0644*	-0.0782*	$0.1541^{*}$	0.0185	0.2307*	0.2909*	$0.2730^{*}$
	NS	-0.1256*	-0.1140*	$0.1471^{*}$	0.0703*	0.1879*	$0.3615^{*}$	0.2347*
CF	Canada	$0.0780^{*}$	$0.1024^{*}$	$0.2871^{*}$	-0.0222	$0.1726^{*}$	0.0667*	$0.2154^{*}$
	SU	$0.0764^{*}$	$0.0521^{*}$	$0.1385^{*}$	$0.0685^{*}$	0.0227	-0.1049*	$0.1212^{*}$
BTM	Canada	$0.2514^{*}$	$0.2146^{*}$	$0.2656^{*}$	-0.1486*	$0.1802^{*}$	-0.0893*	0.0407
	NS	$0.1031^{*}$	0.1245*	$0.1641^{*}$	-0.1920*	$0.1001^{*}$	$-0.0386^{*}$	$0.0416^{*}$
SALES GROWTH	Canada	0.0021	0.0272	-0.0003	-0.0206	-0.0541*	-0.0563*	0.0095
	NS	-0.0031	0.0065	-0.0189	0.0320	$-0.0431^{*}$	-0.0560*	-0.0132
STIR	Canada	-0.0500*	-0.0396	-0.0212	-0.1072*	-0.0379	0.0723*	0.0330
	NS	0.0410*	$0.0918^{*}$	$0.1451^{*}$	-0.0965*	$0.1141^{*}$	$0.1290^{*}$	$0.1456^{*}$
INTERNATIONAL	Canada	$0.0734^{*}$	$0.0958^{*}$	$0.0834^{*}$	$0.0604^{*}$	-0.0024	-0.0638*	$0.0922^{*}$
	NS	$0.1329^{*}$	$0.1196^{*}$	$0.0385^{*}$	0.0742*	0.0223	-0.0501*	$0.0610^{*}$
Note: The table reports the	he Spearman ra	ink correlation	between the	variables for (	Canadian and I	J.S. samples. ]	Note that * shc	ws that the

TABLE 3. (Continued)

correlations are significant at the 5% level.

VariableIIIIII $CANADA$ $-49.6765^{***}$ $-28.4566^{***}$ $-0.0571^{***}$ $CANADA$ $-49.6765^{***}$ $-28.4566^{***}$ $-0.0571^{***}$ $SIZE$ $0.7945$ $-1.9618^{***}$ $-0.0137^{***}$ $0.7945$ $-1.9618^{***}$ $-0.0137^{***}$ $0.7945$ $-1.9618^{***}$ $-0.0137^{***}$ $0.677$ $(-3.79)$ $(-9.26)$ $0.677$ $(-3.35)$ $(-11.22)$ $0.677$ $(-3.35)$ $(-18.47)$ $0.677$ $(-3.35)$ $(-18.47)$ $0.677$ $(-2.72)$ $(-3.35)$ $(-18.47)$ $(-2.272)$ $(-3.35)$ $(-18.47)$ $(-2.272)$ $(-18.47)$ $0.0011^{***}$ $(-2.72)$ $(-4.08)$ $0.011^{***}$ $(-2.73)$ $(-1.8.47)$ $0.011^{***}$ $(-2.73)$ $(-1.8.47)$ $0.011^{***}$ $(-2.45)$ $(-6.85)$ $0.011^{***}$ $(-2.45)$ $(-6.85)$ $0.0011^{***}$ $(-2.45)$ $(-6.85)$ $0.0011^{***}$ $(-2.45)$ $(-3.34)$ $0.011^{***}$ $(-2.45)$ $(-2.98)$ $0.0011^{***}$ $(-2.45)$ $(-0.61)$ $0.180$ $(-2.24)$ $(-2.98)$ $0.001$ $(-2.24)$ $(-2.98)$ $0.006$ $(-2.45)$ $(-0.61)$ $0.180$ $(-2.45)$ $(-0.61)$ $0.180$ $(-2.45)$ $(-2.98)$ $0.180$ $(-2.24)$ $(-2.98)$ $0.180$ $(-2.98)$ $(-0.61)$ $0.180$ $(-2.98)$ $(-0.61)$ <th>MAN W</th> <th></th> <th>SM</th> <th>ß</th>	MAN W		SM	ß
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Ι	II	Ι	II
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	-0.0571*** -0.	.0477***	$-0.0622^{***}$	-0.0425 ***
SIZE $0.7945$ $-1.9618***$ $-0.0137***$ 0.67 $(-3.79)$ $(-3.79)$ $(-9.26)INTANGIBLES -35.5503*** -17.6551*** -0.2422***(-2.72)$ $(-3.35)$ $(-4.03)$ $(-9.26)LEVERAGE -57.3363*** -17.3045*** -0.2422***(-5.55)$ $(-4.08)$ $(-6.85)-0.1301$ $0.3193**$ $(-0.7933***)R&D -0.1301 0.3193** (-0.7933***)(-0.71)$ $(-2.45)$ $(-6.85)$ $(-6.85)-0.0068(-0.61)Risk - CFV -2.6736*** -1.6789*** -0.0066(-0.61)RISE CSH RESET GROWTH -2.6736** -1.6789*** -0.0068(-0.61)FIXED ASSET GROWTH -2.6736** -1.6789*** -0.0068(-0.61)SALES VOLATILITY -58.1469*** -35.6076*** -0.3490***(-6.42)$ $(-11.31)$ $(-3.27)$	(-11.22) (-12.	.21)	(-8.74)	(-11.43)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	-0.0137*** -0.	$.0166^{***}$	-0.0008	-0.0055***
INTANGIBLES -35.5503*** -17.6551*** -0.2422*** LEVERAGE -3.35) (-18.47) (-2.72) (-3.35) (-18.47) -57.3363*** -17.3045*** -0.0793*** (-5.55) (-4.08) (-6.85) -0.1301 0.3193** (-6.85) -0.0011*** (-0.42) (2.45) (-0.011*** (-0.42) (2.45) (-0.011*** (-0.77) (-1.677) (-0.38) (-5.84) Risk - CFV (-2.34) (-2.98) (-0.61) FIXED ASSET GROWTH (-2.24) (-2.98) (-0.61) CASH RESERVE (-6.42) (-1.49) (-3.27) SALES VOLATILITY (-5.81469*** -40.5247*** -0.0390***	(-14.	(86)	(-0.44)	(-5.84)
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	-0.2422*** -0.	.2221***	$-0.1213^{***}$	$-0.0431^{***}$
LEVERAGE -57.3363*** -17.3045*** -0.0793*** AGE -5.55 (-5.55) (-4.08) (-6.85) -0.1301 0.3193** 0.0011*** (-0.42) (2.45) (-6.85) R&D (-0.42) (2.45) (-0.51] R&D (-0.97) (0.38) (-5.84) -16.7871 4.9150 -0.0512*** (-0.97) (0.38) (-5.84) -1.6789*** -0.0006 (-2.34) (-2.98) (-0.61) FIXED ASSET GROWTH 2.2522 1.0967 -0.0068 (-2.34) (-2.98) (-0.61) -2.654428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-1.49) SALES VOLATILITY -58.1469*** -0.3309*** (-6.42) (-11.31) (-3.27)	(-18.47) (-21.	.26)	(-5.29)	(-4.33)
AGE $(-5.55)$ $(-4.08)$ $(-6.85)$ AGE $-0.1301$ $0.3193**$ $0.0011***$ R&D $(-0.42)$ $(2.45)$ $(2.97)$ R&D $(-0.42)$ $(2.45)$ $(2.97)$ R&L $(-0.97)$ $(0.38)$ $(-5.84)$ Risk - CFV $(-0.97)$ $(0.38)$ $(-5.84)$ FIXED ASSET GROWTH $-2.6736**$ $-1.6789***$ $-0.0068$ REXED $(-0.97)$ $(0.38)$ $(-5.84)$ Risk - CFV $(-2.34)$ $(-2.98)$ $(-0.61)$ RISED ASSET GROWTH $2.2522$ $1.0967$ $-0.0068$ CASH RESERVE $-65.4428***$ $-35.6076***$ $-0.3490***$ SALES VOLATILITY $-58.1469***$ $-40.5247***$ $-0.0390***$ C-65.42 $(-11.31)$ $(-3.27)$	-0.0793*** -0	$.0661^{***}$	-0.0685***	-0.0394***
AGE -0.1301 0.3193** 0.0011*** (-0.42) (2.45) (2.97) R&D -16.7871 4.9150 -0.0512*** (-0.97) (0.38) (-5.84) Risk - CFV -2.6736** -1.6789*** -0.0006 (-2.34) (-2.98) (-0.61) FIXED ASSET GROWTH 2.2522 1.0967 -0.0668 (-2.34) (-2.98) (-0.61) CASH RESERVE -5.4428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-1.49) SALES VOLATILITY -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	(-6.85) (-7	.04)	(-4.04)	(-5.32)
R&D         (-0.42)         (2.45)         (2.97)           R&D         -16.7871         4.9150         -0.0512***           Risk - CFV         (-0.97)         (0.38)         (-5.84)           Risk - CFV         (-2.6736**         -1.6789***         -0.0006           FIXED ASSET GROWTH         2.2572         1.0967         -0.0068           CASH RESERVE         (-2.34)         (-2.98)         (-0.61)           CASH RESERVE         -65.4428***         -35.6076***         -0.3490***           SALES VOLATILITY         -58.1469***         -40.5247***         -0.3390***	0.0011*** 0	$.0013^{***}$	-0.0003	0.0007***
R&D -16.7871 4.9150 -0.0512*** Risk - CFV (-0.97) (0.38) (-5.84) -2.6736** -1.6789*** -0.0006 (-2.34) (-2.98) (-0.61) FIXED ASSET GROWTH 2.2522 1.0967 -0.0068 (0.48) (0.55) (-1.49) CASH RESERVE -55.4428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-1.49) SALES VOLATILITY -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	(2.97) (4	.67)	(-0.54)	(2.72)
Risk – CFV (-0.97) (0.38) (-5.84) –2.6736** –1.6789*** –0.0006 (-2.34) (-2.98) (-0.61) FIXED ASSET GROWTH 2.2522 1.0967 –0.0068 (0.48) (0.55) (-1.49) CASH RESERVE –55.4428*** –35.6076*** –0.3490*** (-4.64) (-4.92) (-23.88) SALES VOLATILITY –58.1469*** –40.5247*** –0.0390*** (-6.42) (-11.31) (-3.27)	$-0.0512^{***}$ $-0.0$	$.0513^{***}$	-0.0325	0.0194
Risk – CFV – 2.6736** –1.6789*** –0.0006 FIXED ASSET GROWTH –2.24) (–2.98) (–0.61) (–2.34) (–2.98) (–0.61) (–0.0068 (–1.49) (0.55) (–1.49) (0.48) (0.55) (–1.49) (–1.49) (–1.49) CASH RESERVE –65.4428*** –35.6076*** –0.3490*** (–4.64) (–4.92) (–1.338) SALES VOLATILITY –58.1469*** –40.5247*** –0.0390***	(-5.84) (-4.	.65)	(-1.11)	(0.60)
FIXED ASSET GROWTH (-2.34) (-2.98) (-0.61) 2.2522 1.0967 -0.0068 (0.48) (0.55) (-1.49) CASH RESERVE -65.4428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-23.88) SALES VOLATILITY -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	-0.0006 -0	.0010	-0.0011	$-0.0028^{***}$
FIXED ASSET GROWTH 2.2522 1.0967 -0.0068 (0.48) (0.55) (-1.49) CASH RESERVE -65.4428** -35.6076** -0.3490*** (-4.64) (-4.92) (-23.88) SALES VOLATILITY -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	(-0.61) $(-0.61)$	.94)	(-0.66)	(-2.78)
CASH RESERVE (0.48) (0.55) (-1.49) -65.4428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-23.88) -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	-0.0068 -0	$.0084^{**}$	0.0007	0.0057
CASH RESERVE -65.4428*** -35.6076*** -0.3490*** (-4.64) (-4.92) (-23.88) -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	(-1.49) (-2	.25)	(0.0)	(1.34)
(-4.64)       (-4.92)       (-23.88)         SALES VOLATILITY       -58.1469***       -40.5247***       -0.0390***         (-6.42)       (-11.31)       (-3.27)	-0.3490*** -0	.3205***	$-0.1594^{***}$	-0.1335 ***
SALES VOLATILITY -58.1469*** -40.5247*** -0.0390*** (-6.42) (-11.31) (-3.27)	(-23.88) (-31.	.65)	(-6.66)	(-10.51)
(-6.42) $(-11.31)$ $(-3.27)$	-0.0390*** -0	.0008	$-0.1246^{***}$	$-0.0902^{***}$
	(-3.27) $(-0.5)$	.08)	(-8.48)	(-12.24)
CF 1.3780 1.9239*** 0.0031***	$0.0031^{***}$ 0	$.0034^{***}$	0.0021	0.0029***
(1.46) $(3.37)$ $(3.86)$	(3.86) (3.	.84)	(1.38)	(2.84)

TABLE 4. Difference in Working Capital Policies – Canada Versus U.S.

	CC	C	MA	R	MS	R
Variable	Ι	II	I	II	Ι	II
BTM	12.5683***	$5.8046^{***}$	0.0123***	$0.0047^{**}$	$0.0214^{***}$	0.0126***
	(6.41)	(7.73)	(4.73)	(2.53)	(6.11)	(11.58)
SALES GROWTH	-3.0227	-1.5247	-0.0007	0.0014	0.0068	0.0043
	(-0.43)	(-0.65)	(-0.12)	(0.30)	(0.53)	(1.00)
STIR	0.3929	2.3797	0.0033	0.0018	-0.0007	0.0021
	(0.13)	(1.55)	(0.87)	(0.75)	(-0.14)	(0.89)
INTERNATIONAL	-7.7420**	0.6790	0.0066	0.0115***	$-0.0151^{***}$	0.0023
	(-2.09)	(0.36)	(1.37)	(3.15)	(-2.64)	(0.64)
CONSTANT	$192.5168^{***}$	209.2676***	0.3462***	0.4524***	0.4894***	0.4476***
	(3.75)	(5.82)	(5.22)	(3.92)	(6.48)	(9.45)
YEAR FIXED EFFECTS	Yes	Yes	Yes	Yes	Yes	Yes
INDUSTRY FIXED EFFECTS	Yes	Yes	Yes	Yes	Yes	Yes
FIRM LEVEL CLUSTERING	Yes	Yes	Yes	Yes	Yes	Yes
Adj./Pseudo R-Squared	0.2532	0.1919	0.4948	0.3377	0.2545	0.2051
Z	4477	4477	4477	4477	4477	4477
<b>Note:</b> The regression results are industry. The variable definitions a	based on matched re provided in tabl	samples of Canadi e 1. The depender	an and U.S. firms f it variables are the	rom 1988 to 2016 e three alternative	using PSM based of measures of the v	on firm size and vorking capital

TABLE 4. (Continued)

Working Capital Investment: A Comparative Study – Canada vs US

management. The estimation uses the pooled sample of Canadian and U.S. firms and includes country dummy *CANADA* as main explanatory variable, which equals 1 if the observation belongs to the Canadian sample and 0 otherwise. Specifications in Model I use OLS Regressions and specifications in Model II use Quintile/Median Regressions. All specifications control for year and industry fixed effects and firm-level clustering. The results report the coefficients, t-statistics (in parentheses), adjusted (OLS Models) or Pseudo (Median Models) R-Squared, and the number of observations. Note: \*\*\*, \*\*, \*\* show statistical significance at 1%, 5%, and 10% levels, respectively.

#### V. Determinants of working capital differences

With working capital investment differing significantly between Canada and the U.S., the reasons for the difference are explored. As noted earlier, the article argues that optimal investment in working capital is a trade-off between the cost of financing it and the benefit of investing in it. Consequently, the study explores two possible channels of working capital investment differences between Canada and the U.S. One, the smaller size of the Canadian economy relative to the U.S. economy generates a much smaller aggregate supply of short-term capital to finance operations, increasing the average short-term cost of capital in Canada. This argument is partially supported in the univariate comparison, which shows that compared to the U.S., the average Canadian short-term interest rate (STIR) is 17 basis points higher. The study predicts that higher short-term interest rates in Canada contribute to the difference in working capital investment between Canadian and U.S. firms. A higher short-term interest rate typically increases the cost of, for example, the revolving lines of credit typically used to finance working capital (Aktas, Croci and Petmezas, 2015). It is therefore anticipated that higher short-term interest rates negatively influence working capital investment in Canada. Second, the study suggests as a second channel for lower investment in working capital the relatively lower dependence of Canadian vis-à-vis U.S. firms on foreign sales. Borrowing from Albuquerque, Ramadorai and Watugala (2015) who assessed the extent of the trade credit effect on a firm's stock return based on its international sales exposure, this study uses the international operations dummy as a proxy for foreign sales. Albuquerque, Ramadorai and Watugala (2015) do not report the effect of foreign sales on trade credit (an independent measure of working capital investment). Indeed, it is difficult a priori to anticipate the effect of international operations on working capital investment. The nature of international operation may increase or decrease working capital investment. For instance, an international firm, as opposed to a similar firm with only domestic operations, needs to maintain relatively higher inventories in the home country to accommodate longer shipping times for its goods to foreign customers. Likewise, it may take longer for an international firm to collect receivables from a foreign customer, resulting in a higher working capital investment in receivables. Even if the firm offers the same credit terms to international and domestic customers, the extended transit time for international shipping will delay the start-time of the credit period. On the other hand, an international firm may be able to finance trade credit in the foreign country at a lower cost and/or produce or store finish goods inventory in a foreign country close to its foreign customers to reduce investment in inventory. Such

90

strategies may result in a lower investment in trade credit and in the cash conversion cycle. However, since the univariate analysis shows that compared to the U.S., fewer firms in the Canadian sample have international operations (34% and 7%, respectively), the study predicts that international operations increase working capital investment.

With respect to the empirical methodology, the following model is expected to determine the effect of short-term interest rates and international operation on the difference between the working capital investment of Canadian and U.S. firms.

$$WCM_{it} = \gamma_0 + \gamma_1 CANADA + \gamma_2 CANADA * STIR_{jt}$$

$$or INTERNATIONAL_{it} + \gamma_3 CONTROLS_{it} + \varepsilon_{it}$$
(2)

where  $WCM_{ii}$ , CANADA and  $CONTROLS_{ii}$  are as defined above.  $STIR_{ji}$  is the short-term interest rate in year t for country j, and  $INTERNATIONAL_{ii}$  is a dummy for international operations, which equals 1 if firm i reports foreign income in year t and 0 otherwise, and  $\gamma_i$  and  $\varepsilon_{ii}$  are model parameters. Here,  $\gamma_2$ , the coefficient for the interaction term between the *CANADA* dummy and short-term interest rates/international dummy is of interest, and a significant value of this parameter will confirm if the working capital difference between Canada and the U.S. is affected by short-term interest rates and/or international operations. In particular, a negative (positive) and significant  $\gamma_2$  coefficient will support the argument that short-term interest rates (international operations) decrease (increase) the working capital investment of Canadian firms. The results are reported in table 5.

The findings for mean and median regressions shown in Panels I and II, respectively, support both arguments. The interaction term between short-term interest rates and the three proxies for working capital investment are negative and generally significant, except for the CCC measure, which is insignificant. These results provide support for the conjecture that higher short-term interest rates lead to the lower working capital investment. For the international dummy, the interaction terms for all three proxies of working capital investment are positive and significant at the 1% level. This finding confirms that working capital investment is influenced by the international status of industrial firms. Given that, compared to the U.S., a smaller proportion of Canadian firms have international operations, the study concludes that Canadian firms are able to maintain a lower working capital investment than their U.S. counterparts.

		-				
	CC	CC	M	AR	MS	SR
Variable	Ι	II	Ι	II	Ι	Ш
CANADA	-50.2419***	-29.3287***	-0.0517***	$-0.0392^{***}$	$-0.0608^{***}$	-0.0343 ***
	(-8.52)	(-10.01)	(-7.39)	(-7.47)	(-6.28)	(-6.80)
STIR	0.8841	1.9141	0.0048	0.0026	0.0005	0.0011
	(0.28)	(1.24)	(1.27)	(1.08)	(0.10)	(0.41)
STIR * CANADA	-1.1390	-0.7273	$-0.0046^{***}$	$-0.0064^{***}$	-0.0034	-0.0056***
	(-0.75)	(-0.92)	(-2.61)	(-4.62)	(-1.32)	(-4.23)
INTERNATIONAL	-12.1089***	-3.1823*	-0.0007	0.0043	-0.0238***	-0.0057
	(-3.03)	(-1.67)	(-0.13)	(1.00)	(-3.76)	(-1.61)
INTERNATIONAL * CANADA	26.6658***	22.0091***	$0.0466^{***}$	0.0483 * * *	0.0544 * * *	$0.0496^{***}$
	(3.01)	(3.40)	(4.22)	(5.65)	(4.44)	(5.66)
SIZE	0.8830	$-2.0984^{***}$	-0.0134***	$-0.0160^{***}$	-0.0006	-0.0058***
	(0.75)	(-4.46)	(-9.11)	(-14.29)	(-0.32)	(-6.33)
INTANGIBLES	-34.7789***	$-15.1064^{***}$	-0.2409***	-0.2178***	-0.1197 ***	-0.0445***
	(-2.66)	(-4.07)	(-18.37)	(-20.36)	(-5.23)	(-4.46)
LEVERAGE	$-57.7316^{***}$	-20.5534***	-0.0802 * * *	$-0.0661^{***}$	$-0.0694^{***}$	-0.0399***
	(-5.59)	(-5.12)	(-6.94)	(-6.44)	(-4.09)	(-4.95)
AGE	-0.1412	0.2963**	$0.0010^{***}$	$0.0010^{***}$	-0.0003	0.0008***
	(-0.45)	(2.40)	(2.81)	(3.49)	(-0.62)	(3.32)
R&D	-16.3824	1.4861	-0.0505 ***	-0.0520 * * *	-0.0317	0.0232
	(-0.95)	(0.11)	(-5.77)	(-4.53)	(-1.08)	(0.73)
Risk - CFV	-2.7457**	$-1.6218^{***}$	-0.0007	-0.0013	-0.0013	$-0.0026^{***}$
	(-2.40)	(-2.60)	(-0.76)	(-1.16)	(-0.75)	(-2.69)
FIXED ASSET GROWTH	2.0569	0.9353	-0.0071	$-0.0084^{**}$	0.0003	0.0078*
	(-2.40)	(-2.60)	(-0.76)	(-1.16)	(-0.75)	(-2.69)
		( Contin	( pənı			

TABLE 5. Determinants of Difference in Working Capital Policies – Canada Versus U.S.

	G	c	M.∤	AR	SM	R
Variable	Ι	Π	I	II	Ι	Π
CASH RESERVE	$-65.0401^{***}$	$-37.3080^{***}$	-0.3479***	$-0.3180^{***}$	$-0.1584^{***}$	$-0.1398^{***}$
	(-4.62)	(-5.55)	(-23.83)	(-28.58)	(-6.62)	(-11.53)
SALES VOLATILITY	-58.0464***	-41.3820***	-0.0385***	0.0060	-0.1243***	-0.0920***
	(-6.41)	(-14.93)	(-3.23)	(0.71)	(-8.47)	(-11.65)
CF	1.3760	1.7334***	$0.0031^{***}$	$0.0031^{***}$	0.0021	$0.0030^{***}$
	(1.45)	(2.97)	(3.85)	(3.76)	(1.38)	(2.89)
BTM	12.4477***	6.5942***	$0.0120^{***}$	0.0043 **	0.0212***	$0.0114^{***}$
	(6.35)	(10.48)	(4.63)	(2.43)	(6.03)	(10.32)
SALES GROWTH	-3.4426	-2.9676*	-0.0017	0.0002	0.0058	0.0022
	(-0.48)	(-1.89)	(-0.28)	(0.07)	(0.46)	(0.52)
CONSTANT	194.3455***	220.6769***	0.3533***	0.4803***	0.4947***	0.4639***
	(3.76)	(5.68)	(5.34)	(4.13)	(6.55)	(7.12)
YEAR FIXED EFFECTS	Yes	Yes	Yes	Yes	Yes	Yes
INDUSTRY FIXED EFFECTS	Yes	Yes	Yes	Yes	Yes	Yes
FIRM LEVEL CLUSTERING	Yes	Yes	Yes	Yes	Yes	Yes
Adj./Pseudo R-Squared	0.2540	0.1928	0.4969	0.3409	0.2562	0.2074
N	4477	4477	4477	4477	4477	4477
Note: The regression results are industry. The variable definitions an management. The estimation uses th	s based on matched re provided in tabl re pooled sample o	samples of Canad le 1. The depende of Canadian and U	ian and U.S. firms nt variables are th .S. firms and inclu	from 1988 to 2016 three alternative des country dumm	using PSM based measures of the y CANADA as m	on firm size and working capital ain explanatory

(Continued)	
<b>TABLE 5.</b>	

specifications in Model II use Quintile/Median Regressions. All specifications control for year and industry fixed effects and firm-level clustering. The results report the coefficients, t-statistics (in parentheses), adjusted (OLS Models) or Pseudo (Median Models) R-Squared, and the number of observations. Note: \*\*\*, \*\*, \* show statistical significance at 1%, 5%, and 10% levels, respectively. variable, which equals 1 if the observation belongs to the Canadian sample and 0 otherwise. Specifications in Model I use OLS Regressions and

#### Working Capital Investment: A Comparative Study – Canada vs US

																Μ	ult	ina	atio	ona	l F	'in a	anc	e J	our
	WSR	II	-0.0245***	(-6.00)	-0.0057 ***	(-5.86)	$-0.0250^{***}$	(-3.01)	-0.0403***	(-5.64)	$0.0008^{***}$	(3.09)	0.0227	(0.73)	$-0.0026^{***}$	(-2.68)	0.0050	(1.11)	-0.1327 ***	(-12.58)	-0.0885 ***	(-11.16)	0.0028***	(3.27)	
	Adj-V	Ι	-0.0343 * * *	(-4.85)	-0.0009	(-0.49)	-0.1145***	(-5.03)	$-0.0608^{***}$	(-3.66)	-0.0002	(-0.40)	-0.0211	(-0.74)	-0.0009	(-0.54)	0.0018	(0.23)	-0.1550 ***	(-6.55)	-0.1260 ***	(-8.67)	0.0020	(1.33)	
	WAR	II	-0.0063*	(-1.71)	$-0.0172^{***}$	(-16.50)	-0.1997***	(-20.82)	-0.0589***	(-6.86)	$0.0013^{***}$	(4.83)	$-0.0451^{***}$	(-6.09)	-0.0015	(-1.63)	$-0.0066^{**}$	(-2.44)	-0.3197 ***	(-31.53)	0.0032	(0.33)	$0.0032^{***}$	(3.85)	
	Adj-V	I	-0.0099*	(-1.96)	$-0.0132^{***}$	(-9.02)	-0.2355***	(-17.70)	$-0.0661^{***}$	(-5.79)	0.0012***	(3.25)	$-0.0401^{***}$	(-4.65)	-0.0004	(-0.45)	-0.0060	(-1.32)	$-0.3448^{***}$	(-23.50)	-0.0409 ***	(-3.53)	$0.0030^{***}$	(3.77)	( pən
1	CC	Π	-8.1865***	(-3.93)	$-2.0413^{***}$	(-3.83)	$-11.3101^{**}$	(-2.20)	-19.7098***	(-4.53)	0.4022***	(2.93)	7.2849	(0.58)	$-1.8191^{***}$	(-2.98)	1.4660	(0.75)	-35.9665***	(-4.95)	-37.7587***	(-11.24)	$1.9109^{***}$	(4.12)	( Contin
	Adj-C	I	-20.2759***	(-4.68)	0.8564	(0.74)	$-30.1710^{**}$	(-2.34)	-51.7746***	(-5.16)	-0.0392	(-0.13)	-7.5043	(-0.45)	-2.4521**	(-2.21)	2.8963	(0.63)	$-64.2758^{***}$	(-4.64)	$-57.7626^{***}$	(-6.53)	1.3343	(1.43)	
		Variable	CANADA		SIZE		INTANGIBLES		LEVERAGE		AGE		R&D		Risk - CFV		FIXED ASSET GROWTH		CASH RESERVE		SALES VOLATILITY		CF		

TABLE 6. Difference in Industry-Adjusted Working Capital Policies – Canada Versus U.S.

Adj-C	CCC	Adj-W	/AR	Adj-V	VSR
I	Π	I	Π	I	Π
12.8263***	7.1758***	$0.0133^{***}$	$0.0054^{***}$	0.0220 * * *	0.0125***
(6.65)	(12.04)	(5.12)	(3.61)	(6.31)	(10.39)
-2.7561	-0.5562	-0.0006	0.0003	0.0069	0.0042
(-0.40)	(-0.25)	(-0.10)	(0.08)	(0.55)	(1.21)
0.9018	0.8247	0.0044	0.0018	-0.0000	0.0033
(0.29)	(0.59)	(1.16)	(0.64)	(-0.01)	(1.09)
-6.3032*	1.7238	0.0096**	$0.0178^{***}$	$-0.0122^{**}$	0.0018
(-1.73)	(0.00)	(2.01)	(5.01)	(-2.17)	(0.50)
9.5152	51.6335	0.1034	0.3052**	0.1026	0.0569*
(0.19)	(1.47)	(1.55)	(2.22)	(1.38)	(1.69)
Yes	Yes	Yes	Yes	Yes	Yes
Yes	Yes	Yes	Yes	Yes	Yes
Yes	Yes	Yes	Yes	Yes	Yes
0.1010	0.047	0.2764	0.173	0.1010	0.061
4477	4477	4477	4477	4477	4477
based on matched e provided in table	samples of Canadi 1. The dependent	an and U.S. firms f variables are adju	rom 1988 to 2016 isted values of the	using PSM based of three alternative r	n firm size and neasures of the
	Adj-C I Adj-C Adj-C Adj-C (6.65) -2.7561 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-0.40) 0.9018 (-1.73) 9.5152 (-1.7552 (-	Adj-CCC           I         I           12.8263***         7.1758***           12.8263***         7.1758***           (6.65)         (12.04)           -2.7561         -0.5562           (-0.40)         0.8247           (0.29)         0.8247           (0.29)         0.8247           (12.73)         0.8247           (0.29)         0.8247           (0.19)         (1.47)           Yes         Yes           Yes         Yes           Yes         Yes           0.1010         0.047           absed on matched samples of Canadii           eprovided in table 1. The dependent	Adj-CCC         Adj-W           I         I         I  <	Adj-CCCAdj-WARIIIIIIIIIIIII2.8263***7.1758*** $0.0054***$ (6.65)(12.04)(5.12) $0.0054***$ (6.65)(12.04)(5.12) $0.0054***$ (6.65)(12.04)(5.12) $0.00354***$ (0.40) $-0.5562$ $-0.00066$ $0.0033$ (-0.40)(-0.25) $(-0.10)$ $0.088$ (0.29) $0.9018$ $0.8247$ $0.0044$ $0.0018$ (0.29) $(0.59)$ $(1.16)$ $(0.64)$ -6.3032** $1.7238$ $0.0096**$ $0.0178***$ (-1.73) $(0.90)$ $(2.01)$ $(0.178***)$ (-1.73) $(0.90)$ $(2.01)$ $(5.01)$ 9.5152 $51.6335$ $0.1034$ $0.3052**$ $(0.19)$ $(1.47)$ $(1.55)$ $(2.22)$ YesYesYesYesYesYesYesYesYesYesYesYes $4477$ $4477$ $4477$ $4477$ $e$ provided in table 1. The dependent variables are adjusted values of the $e$ provided in table 1. The dependent variables are adjusted values of the	Adj-CCC         Adj-WAR         Adj-VAR         Adj-VAR         Adj-VAR         Adj-V           1         1         1         1         1         1         1         Adj-VAR         0.0220***         0.0220***         0.0220***         0.0059         0.0006         0.0006         0.0006         0.0006         0.0018         0.0019         0.0172***         0.1026         0.0010         0.0122***         0.1026         0.1026         0.1026         0.1026         0.1026         0.1026         0.1026         0.1026         0.1026         0.1026         Yes         Yes

TABLE 6. (Continued)

explanatory variable, which equals 1 if the observation belongs to the Canadian sample and 0 otherwise. Specifications in Model I use OLS Regressions with adjusted dependent variables actual minus the mean industry value of the dependent variables) and specifications in Model I use QLS Quintile/Median Regressions with adjusted dependent variables (actual minus the median industry value of the dependent variables). All specifications control for year and industry fixed effects and firm level clustering. The results report the coefficients, t-statistics (in parentheses), adjusted (OLS Models) or Pseudo (Median Models) R-Squared, and the number of observations. Note: \*\*\*, \*\*, \* show statistical significance at 1%, 5%, and 10% levels, respectively.

Working Capital Investment: A Comparative Study – Canada vs US

#### VI. Additional results and robustness checks

This section discusses additional tests to ensure the robustness of the findings.<sup>11</sup> First, the current specification assumes that industry effects are similar between Canada and the U.S. This is a plausible assumption given the highly integrated supply chain between the two countries. Nevertheless, to address this concern industry-adjusted values of the dependent variables are used. Specifically, the three working capital measures are adjusted by subtracting mean (median) industry (using 2-digit SIC code) value from the actual value and then running mean (median) regressions. The results reported in table 6 are similar to those in table 4 as there is a negative and significant *CANADA* dummy for mean and median regressions.

Second, since the multivariable specifications in this study include a large number of control variables, it is important to study whether the explanatory variables are robust. Learner's (1985) extreme bounds analysis (EBA) mitigates this concern with a 95% confidence interval. EBA estimates extreme values of the coefficient of interest using all possible linear combinations regressions. The estimated coefficient of the explanatory variable being tested is robust if it falls within the EBA estimated extreme values (minimum and maximum coefficients) and does not change sign at the extreme values. The analysis shows that extreme bounds for three variables (AGE, GFA and SALES GROWTH) do not exist across all three working capital measures. In addition, four variables (INTANGIBLES, RISK-CFV, R&D and CASH RESERVE) are fragile because analysis does not find extreme bounds for at least one of the working capital measures. The remaining explanatory variables are found to be robust. Consequently, the study re-estimates Specification 1 twice by excluding three and seven fragile variables. The findings are quantitatively and qualitatively very similar to those reported in table 4.

Third, an alternative measure is used for the cash conversion cycle. Consistent with most of the literature, this cycle is calculated using the average of the beginning and ending values of working capital accounts, such as inventory, receivables and payables. In an alternative measure, the cash conversion cycle is computed using only the year-end values of the working capital accounts and repeated for the estimation of Models 1 and 2. The results are very similar to the findings reported in tables 4 and 5.

Fourth, instead of the matched sample with the PSM technique, the

96

<sup>11.</sup> The results of these robustness tests, except for the first one, are not reported here but are available upon request.

original raw sample uses 9,602 and 67,820 firm-year observations for Canada and U.S., respectively, and repeats the complete analysis. Again, the results are qualitatively and quantitatively similar to the findings from the matched sample.

Next, an alternative method would be to match the firms using additional firm characteristics. Therefore, the study uses PSM for firm size, leverage and book-to-market to create a matched sample and re-estimate the complete set of results. Similar results are obtained with this alternative matching strategy. Finally, Baños-Caballero, García-Teruel and Martínez-Solano (2010) and Chiou, Cheng and Wu (2006) report a relationship with mixed findings – between working capital investment and firm-level profitability (return on assets). This study adds profitability as a control and re-estimates Models 1 and 2. The results are once again similar.<sup>12</sup>

#### **VII.** Conclusion

This appears to be the first study to statistically compare working capital policies of Canadian and U.S. firms using a matched sample over a long time period. The study uses the propensity score matching technique - based on firm size and industry - to find a closest U.S. neighbor for each Canadian firm-year observation to compare working capital investment between the two countries. The findings show that, compared to the U.S. sample, Canadian firms have a substantially lower cash conversion cycle, working capital to asset ratio and non-cash working capital to sales ratio. These differences are not only statistically significant but also economically important. The analysis also confirms that the difference in working capital investment is robust to a number of firm, industry and country controls as well as to year and industry fixed effects. The results are robust to a variety of complementary tests. Finally, the study confirms that lower working capital investment in Canada is affected by higher short-term interest rates and smaller international operations.

Accepted by: Prof. P. Theodossiou, PhD, Editor-in-Chief, May 2019

<sup>12.</sup> Cross-country differences are created in working capital measures (defined as the difference between dependent variable values of U.S. and Canadian firms) and are used as dependent variables in multivariable regressions. Two alternative specifications, one including differences in all explanatory variables and the other just with Canadian data for the explanatory variables, find the results (not reported) generally similar to the main findings.

Appendix. Variable defir	litions
Variables	
Dependent variables:	Definitions (Compustat is the source of variables shown in "italics".)
CCC	Cash conversion cycle: cash conversion cycle based on average of beginning and ending values and is the sum of days' sales outstanding (DSO) and days inventory outstanding (DIO) minus the days' payable outstanding (DPO) times 365, where: DSO: average receivables ( <i>rect</i> ) divided by the sales ( <i>sale</i> ) DIO: average inventories ( <i>invt</i> ) divided by the cost of goods sold ( <i>cogs</i> )
WSR	DPO: average payables ( <i>ap</i> ) divided by the purchases ( $Jinvt+cogs$ ) Working capital to sales ratio: non-cash working capital scaled by the total sales ( <i>sales</i> ), where: non-cash working canital is receivables ( <i>rec</i> () thus inventories ( <i>inv</i> ) minus navables ( <i>an</i> )
WAR	Working capital to assets ratio: non-cash working capital scaled by the total assets ( <i>at</i> )
Working capital accounts:	
CASH	Cash: cash and marketable securities divided by total assets ( <i>chelat</i> )
INV	Inventories: inventory to total assets ratio ( <i>ivntlat</i> )
A/R	Receivables: accounts receivable to total assets ratio ( <i>rect/at</i> )
A/P	Payables: accounts payable to total assets ratio (ap/at)
Main Independent Variable:	
CANADA	Country dummy for Canada: indicator variable for Canada, which equals 1 if the firm-year observation belongs to a Canadian firm and 0 if the firm-year observation belongs to a U.S. firm.

riable definitio Vo div

(Continued)

#### **Multinational Finance Journal**

Appendix. (Continued)	
Variables	
Controls:	Definitions (Compustat is the source of variables shown in "italics".)
SIZE INTANGIBLES	Firm size: natural log of book value of the total assets ln ( <i>at</i> ) Intangible assets: intangible assets divided by the total assets ( <i>intan</i> / <i>at</i> )
LEVERAGE AGE	Book leverage: long term debt plus debt in current liabilities divided by total assets (( <i>dlttq+dlcq</i> )/ <i>atq</i> )) Firm age: number of vear since initial public offering on June 1, 2017 ( <i>ipodate</i> )
R&D	Research and development: research and development expenditure to sales ratio (xrd/sale) (set to 0 if xrd is
Risk – CFV	Cash flow volatility: coefficient of variation in cash flow from operations during last 10 years, where: Cash flow is: income before extraordinary items plus depreciation scaled by net property. plant and equipment
	((ib+dp)/ppent)
GROWTH in FIXED ASSET	Fixed asset growth: change in the net property, plant and equipment over the previous fiscal year ((ppent <sub>i-l</sub> -
(GFA) CASH RESERVE	<i>ppent<sub>i-2</sub>)/ppent<sub>i-2</sub>)</i> Cash reserve: ratio of cash and short-term investment to total assets ( <i>chelat</i> )
SALES VOLATILITY	Sales volatility: coefficient of variation in annual sales to total asset ratio over the past 10 years
CF	Cash flows: the sum of income before extraordinary items and depreciation scaled by the net property, plant, and equipment $((ib+dp)/ppent)$
BTM	Book-to-market: total common equity scaled by market capitalization ( $ceq/(prcc_f * csho)$ ), where:
SALES GROWTH	$prcc_f$ is fiscal year-end stock price and $csho$ is common shares outstanding Sales growth: change in sales over the previous fiscal year $((sale_{r-}sale_{r-1}))/sale_{(r-1)})$
STIR	Short-term interest rates: annual percentage interest rates based on 3-month Treasury bill rate
INTERNATIONAL	International dummy: indicator variable that equals 1 if firm reports a positive pre-tax foreign income and 0 otherwise (if $pifo>0$ )
	× ×

#### References

- Anagnostopoulou, S. 2013. Working capital management and firm listing status. *Multinational Finance Journal* 16(3-4): 261-301.
- Aktas, N.; Croci, E.; and Petmezas, D. 2015. Is working capital management value-enhancing? Evidence from firm performance and investments. *Journal of Corporate Finance* 30: 98-113.
- Albuquerque, R.; Ramadorai, T.; and Watugala, S. W. 2015. Trade credit and cross-country predictable firm returns. *Journal of Financial Economics* 115(3): 592-613.
- Almeida, J. R., and Eid, W. 2014. Access to finance, working capital management and company value: Evidences from Brazilian companies listed on BM&FBOVESPA. *Journal of Business Research* 67(5): 924-934.
- Baker, K.; Dutta, S.; and Saadi, S. 2011. Corporate finance practices in Canada: Where do we stand? *Multinational Finance Journal* 15(3-4): 157-192.
- Baker, K. H., and Mukherjee, T. K. 2007. Survey research in finance: Views from journal editors. *International Journal of Managerial Finance* 3(1): 11-25.
- Baños-Caballero, S.; García-Teruel, P. J.; and Martínez-Solano, P. 2010. Working capital management in SMEs. *Accounting and Finance* 50(3): 511-527.
- Bates, T. W.; Kahle, K. M.; and Stulz, R. M. 2009. Why do US firms hold so much more cash than they used to? *Journal of Finance* 64(5): 1985-2021.
- Belt, B., and Smith, K. V. 1991. Comparison of working capital management practices in Australia and the United States. *Global Finance Journal* 2(1-2): 27-54.
- Ben-Nasr, H. 2016. State and foreign ownership and the value of working capital management. *Journal of Corporate Finance* 41: 217-240.
- Bhojraj, S.; Lee, C. M.; and Oler, D. K. 2003. What's my line? A comparison of industry classification schemes for capital market research. *Journal of Accounting Research* 41(5): 745-774.
- Bierman, H.; Chopra, K.; and Thomas, J. 1975. Ruin considerations: Optimal working capital and capital structure. *Journal of Financial and Quantitative Analysis* 10(1): 119-128.
- Box, T.; Davis, R. L.; Hill, M. D.; and Lawrey, C. M. 2018. Operating performance and aggressive trade credit policies. *Journal of Banking & Finance* 89: 192-208.
- Chiou, J.; Cheng, L.; and Wu, H. 2006. The determinants of working capital management. *Journal of American Academy of Business* 10(1): 149-155.
- Deloof, M. 2003. Does working capital management affect profitability of Belgian firms? *Journal of Business Finance and Accounting* 30(34): 573-588.
- Ding, S.; Guariglia, A.; and Knight, J. 2013. Investment and financing constraints in China: Does working capital management make a difference?

#### 100

Journal of Banking and Finance 37(5): 1490-1507.

- Eckbo, B. E., and Kisser, M. 2013. Corporate funding: Who finances externally? Working paper. Hanover, NH: Tuck School of Business, Dartmouth College.
- Elo, K., and Tanska, T. 2016. Working capital management and national culture: A cross country study. Master's thesis. Uppsala, Sweden: Department of Business Studies, Uppsala University.
- Fazzari, S. M., and Petersen, B. C. 1993. Working capital and fixed investment: New evidence on financing constraints. *Rand Journal of Economics* 24(3): 328-342.
- García-Teruel, J. P., and Martinez-Solano, P. 2007. Effects of working capital management on SME profitability. *International Journal of Managerial Finance* 3(2): 164-177.
- Ge, Y., and Qiu, J. 2007. Financial development, bank discrimination and trade credit. *Journal of Banking & Finance* 31(2): 513-530.
- Gill, A. S., and Biger, N. 2013. The impact of corporate governance on working capital management efficiency of American manufacturing firms. *Managerial Finance* 39(2): 116-132.
- Graham, J. R. 2004. Roundtable on corporate disclosure. *Journal of Applied Corporate Finance* 16(4): 338-362.
- Hawawini, G.; Viallet, C.; and Vora, A. 1986. Industry influence on corporate working capital decisions. *Sloan Management Review* 27(4): 15-24.
- Khoury, N. T.; Smith, K. V.; and MacKay, P. I. 1999. Comparing working capital practices in Canada, the United States, and Australia: A note. *Canadian Journal of Administrative Sciences* 16(1): 53-57.
- Kieschnick, R.; Laplante, M.; and Moussawi, R. 2013. Working capital management and shareholders' wealth. *Review of Finance* 17(5): 1827-1852.
- Krishnan, J., and Press, E. 2003. The North American industry classification system and its implications for accounting research. *Contemporary Accounting Research* 20(4): 685-717.
- Lazaridis, I., and Tryfonidis, D. 2006. Relationship between working capital management and profitability of listed companies in the Athens stock exchange. *Journal of Financial Management and Analysis* 19(1): 26-35.
- Leamer, E. E. 1985. Sensitivity analyses would help. American Economic Review 75(3): 308-313.
- Over a Barrel. 2016. CFO Magazine (July 14). Retrieved from http://ww2.cfo.com/cash-flow/2016/07/barrel/
- Petersen, M. A., and Rajan, R. G. 1997. Trade credit: Theories and evidence. *Review of Financial Studies* 10(3): 661-691.
- Ramiah, V.; Zhao, Y.; Moosa, I.; and Graham, M. 2016. A behavioural finance approach to working capital management. *European Journal of Finance* 22(8-9): 662-687.
- Ross, S. A.; Westerfield, R. W.; Jordan, B. D.; and Roberts, G. S. 2016.

*Fundamentals of Corporate Finance*, 9<sup>th</sup> ed. Toronto: McGraw-Hill Ryerson.

- Shin, H., and Soenen, L. 1998. Efficiency of working capital management and corporate profitability. *Financial Practice and Education* 8: 37-45.
- Smith, K. V. 1979. *Guide to Working Capital Management*. Columbus, OH: McGraw-Hill.
- Smith, K.V., and Belt, B. 1989. Working capital management in practice: An update. Institute for Research in the Behavioral, Economic, and Management Sciences, Krannert Graduate School of Management, Purdue University.
- Weiner, C. 2005. The impact of industry classification schemes on financial research. SFB 649 discussion paper No. 2005,062. Berlin: SFB 649, Economic Risk.
- Weinraub, H. J., and Visscher, S. 1998. Industry practice relating to aggressive conservative working capital policies. *Journal of Financial and Strategic Decision* 11(2): 11-18.
- Zeidan, R., and Shapir, O. M. 2017. Cash conversion cycle and value-enhancing operations: Theory and evidence for a free lunch. *Journal of Corporate Finance* 45: 203-219.